Author's response to reviews

Title: Use of dietary supplements by cardiologists, dermatologists and orthopedists: report of a survey

Authors:

Annette Dickinson (annettedickinson@comcast.net)
Andrew Shao (ashao@crnusa.org)
Nicolas Boyon (Nicolas.Boyon@Ipsos.com)
Julio C. Franco (Julio.Franco@Ipsos.com)

Version: 5 Date: 23 February 2011

Author's response to reviews:

In this revision, we have corrected a typo in the middle of page 16, to replace the word "reflects" with the word "reflect." We have also inserted a phrase for clarity and to increase conformity with the abstract, in the conclusion paragraph on page 17, where we added the phrase "are related to their specialty and...."